Author’s response to reviews

Title: Clinical Decision Support System for the Management of Osteoporosis Compared to NOGG Guidelines and an Osteology Specialist: A Validation Pilot Study

Authors:
Haukur Gudmundsson (haukurtg@icloud.com)
Karen Hansen (keh@medicine.wisc.edu)
Bjarni Halldorsson (Bjarni.Halldorsson@decode.is)
Bjorn Ludviksson (bjornlud@landspitali.is)
Bjorn Gudbjornsson (bjorngu@landspitali.is)

Version: 1 Date: 19 Dec 2017

Author’s response to reviews:

Dear Editor.

Thank you for your letter. We appreciate your comments and have adjusted our manuscript in accordance to your suggestions:

1) Email addresses are now visible on the title page
2) Objective heading has been removed from the abstract
3) Keywords have been added and now follow the abstract section
4) Introduction heading renamed to “Background”
5) We have updated the information in the "Ethical approval and consent to participate" section of the manuscript. As our study protocol was approved by the Icelandic National Bioethics Committee and the Data Protective Authorities we were not bounden to collect informed consent
6) Authorship criteria has been reviewed, but we made no alterations to the list of authors as all authors to this manuscript have provided significant contribution
7) Figure legends are now at the end of the manuscript
8) Figures have been removed from document and are now uploaded as individual image files (PNG)
We hope that our corrections are satisfactory and we look forward to receiving your reviewer’s feedback.

On behalf of all authors,

Sincerely,

Haukur Tyr Gudmundsson